Author's response to reviews

Title: Chk1 Inhibition as a Novel Therapeutic Strategy for Treating Triple-Negative Breast and Ovarian Cancers

Authors:

Christopher Bryant (C.J.Bryant@bath.ac.uk)
Rebecca Rawlinson (rebecca.rawlinson@bath.edu)
Andrew J Massey (a.massey@vernis.com)

Version: 6 Date: 21 July 2014

Author's response to reviews:

We have added the following sentence to the acknowledgements section detailing the source of funding:

"This work was funded by Vernalis R&D Ltd who approved the manuscript for publication."